Author's response to reviews

Title: Gene Expression Profiling of Laterally Spreading Tumors

Authors:

Shoko Minemura (koshoramunemi@yahoo.co.jp)
Takeshi Tanaka (take197664@yahoo.co.jp)
Makoto Arai (araim-cib@umin.ac.jp)
Kenichiro Okimoto (kenrunaway@yahoo.co.jp)
Arata Oyamada (a-oymada@live.jp)
Keiko Saito (keiko-ueda@y3.dion.ne.jp)
Daisuke Maruoka (d-maruoka@chiba-u.jp)
Tomoaki Matsumura (matsumura919@yahoo.co.jp)
Tomoo Nakagawa (tom20852@yahoo.co.jp)
Tatsuro Katsuno (katsuno@faculty.chiba-u.jp)
Takashi Kishimoto (tkishi@faculty.chiba-u.jp)
Osamu Yokosuka (yokosukao@faculty.chiba-u.jp)

Version: 2 Date: 14 January 2015

Author's response to reviews:

Dear Staffs of Journal Editorial Office BioMed Central

We changed and corrected. We uploaded the revised manuscript.